Author’s response to reviews

Title: Preeclampsia management modalities and perinatal death: a retrospective study in Woldia General Hospital

Authors:

Kenean Tlaye (kenan34@wldu.edu.et)
Melese Endalfer (Melese2235@wldu.edu.et)
Mesfin Kassaw (mesfinkassaw@wldu.edu.et)
Mussie Gebremedhin (mussiemulugeta@gmail.com)
Yared Aynalem (yaredasmare123@dbu.edu.et)

Version: 2 Date: 25 Mar 2020

Author’s response to reviews:

We the authors, would like to thank the reviewer for his constructive comments and unreserved support specially on the language command of the paper, which was one of our biggest challenge. We found that almost all of the comments are important and adjustments has been considered accordingly. We also tried to advance the language command of the paper as much as possible.